Author's response to reviews

Title: Prevalence and progression of visual impairment in patients newly diagnosed with clinical type 2 diabetes: a 6-year follow up study

Authors:

Niels de Fine Olivarius (olivarius@sund.ku.dk)
Volkert Siersma (siersma@sund.ku.dk)
Gitte Juul Almind (gittejuul@gmail.com)
Niels Vesti Nielsen (nvn@dadlnet.dk)

Version: 6 Date: 2 February 2011

Author's response to reviews:

Author list: Niels de Fine Olivarius, Volkert Siersma, Gitte Juul Almind and Niels Vesti Nielsen Title: Prevalence and progression of visual impairment in patients newly diagnosed with clinical type 2 diabetes: a 6-year follow up study

Journal: BMC Public Health

MS ID: 1370621607437611

Dear Prof de Fine Olivarius,

Thank you very much for submitting the revised version of your manuscript. Unfortunately there are still changes that need to be made prior to publication which cannot be made in-house. Please carry out the additional revisions listed below and resubmit your amended manuscript. The instructions to submit your revised manuscript are available below.

We may have edited the manuscript for you, please ensure that all your revisions are made to the most recent version of your manuscript, which can be accessed at the following link:

http://www.biomedcentral.com/media/man/1370621607437611/msd/main/0

Figures can be accessed here:

http://www.biomedcentral.com/media/man/1370621607437611/msd/FIG/1
http://www.biomedcentral.com/media/man/1370621607437611/msd/FIG/2

Please note that this is the final proofing stage. All revised manuscripts submitted to BMC Public Health are regarded as being ready for publication. After publication, you will not be able to make any corrections to errors present in this version. Please check your manuscript carefully for any typing or content errors, as these will be included in the published version.

Please resubmit your manuscript here via the online submission system:
http://www.biomedcentral.com/author/manuscript/details/view.do?manuscriptId=1370621607437611

Formatting changes to be made:
---------------------------------

Major revisions (we require the author to make these changes)

Tables: all tables must be cited in numerical order in the text. Table 5 is currently cited first.
DONE